\documentclass[12pt]{article}

% for graphics
\usepackage{epsfig}

% make figures as follows:
%\begin{figure}
%\begin{center}
%\psfig{figure=eps_file_name}
%\end{center}
%\caption{Put caption here.}
%\label{label}
%\end{figure}

% make units in Roman type with space before
\newcommand{\unit}[1]{\mbox{ #1}}

% centered unnumbered section head
\newcommand{\csection}[1]{\begin{center}{\Large\bf #1}\end{center}}

% make sgn roman
\newcommand{\sgn}{\mbox{sgn}}

% bold math characters
\newcommand{\bold}[1]{\mbox{\bf #1}}
\newcommand{\grbold}[1]{\mbox{\boldmath $#1$}}

% make double space (2) or single space (1)
\renewcommand{\baselinestretch}{1}

% decrease page margins
%\hoffset=-0.4in
%\textwidth=6.2in
%\voffset=-0.5in
%\textheight=8.5in

\begin{document}

\csection{Ph 340 -- Problem Set 7 -- Spring 1997}

\begin{enumerate}
\item
Modify the RLaB program {\tt ho\_demo.r} so that it predicts the
eigenvalues and eigenvectors for a finite square well potential.
Check to see whether it gives the answers you expect from the theory
of the finite square well.
\end{enumerate}

\end{document}
